**Author's response to reviews**

**Title:** The efficacy of dapagliflozin combined with hypoglycemic drugs in treating type 2 diabetes: protocol for meta-analysis of randomized controlled trials

**Authors:**
- Yun-nan Sun (mb258447@umac.mo)
- Yi Zhou (mb25822@umac.mo)
- Xi Chen (mb258447@umac.mo)
- Wen-si Che (mb25822@umac.mo)
- Siu-wai Leung (siu@inf.ed.ac.uk)

**Version:** 5  
**Date:** 30 October 2013

**Author's response to reviews:** see over
Dear Sir/Madam,

**Point-by-point response**

Thank you for your helpful advice and comments for improving the manuscript. We have revised the manuscript according to your advice and comments. The following are our response to your specific comments:

“There is a typo the authors need to address - in the abstract they state “trial registration” and then have the PROSPERO number. It is not a trial register - it is a systematic review protocol register.”

Thanks for your correction. The term “Trial Registration” has been amended to read “Study Registration”.

Yours faithfully,

The authors.